

# G OPEN ACCESS

**Citation:** Vogt L, Reichlin TS, Nathues C, Würbel H (2016) Authorization of Animal Experiments Is Based on Confidence Rather than Evidence of Scientific Rigor. PLoS Biol 14(12): e2000598. doi:10.1371/journal.pbio.2000598

**Academic Editor:** Ulrich Dirnagl, Charité— Universitätsmedizin Berlin, Germany

Received: July 19, 2016

Accepted: November 3, 2016

Published: December 2, 2016

**Copyright:** © 2016 Vogt et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The file containing all raw data from this study is available from the figshare database (https://figshare.com/s/ bc48ed5dff9e6ebd2000) DOI: 10.6084/m9. figshare.4110153.

**Funding:** Swiss Food Safety and Veterinary Office (FSVO) https://www.blv.admin.ch/ (grant number 2.13.01). LV and TSR were fully funded through this grant. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. META-RESEARCH ARTICLE

# Authorization of Animal Experiments Is Based on Confidence Rather than Evidence of Scientific Rigor

#### Lucile Vogt<sup>1</sup>°, Thomas S. Reichlin<sup>1</sup>°, Christina Nathues<sup>2</sup>, Hanno Würbel<sup>1</sup>\*

1 Division of Animal Welfare, Veterinary Public Health Institute, Vetsuisse Faculty, University of Bern, Bern, Switzerland, 2 Division of VPH-Epidemiology, Veterinary Public Health Institute, Vetsuisse Faculty, University of Bern, Liebefeld, Switzerland

These authors contributed equally to this work.

\* hanno.wuerbel@vetsuisse.unibe.ch

# Abstract

Accumulating evidence indicates high risk of bias in preclinical animal research, guestioning the scientific validity and reproducibility of published research findings. Systematic reviews found low rates of reporting of measures against risks of bias in the published literature (e.g., randomization, blinding, sample size calculation) and a correlation between low reporting rates and inflated treatment effects. That most animal research undergoes peer review or ethical review would offer the possibility to detect risks of bias at an earlier stage, before the research has been conducted. For example, in Switzerland, animal experiments are licensed based on a detailed description of the study protocol and a harm-benefit analysis. We therefore screened applications for animal experiments submitted to Swiss authorities (n = 1,277) for the rates at which the use of seven basic measures against bias (allocation concealment, blinding, randomization, sample size calculation, inclusion/exclusion criteria, primary outcome variable, and statistical analysis plan) were described and compared them with the reporting rates of the same measures in a representative sub-sample of publications (n = 50) resulting from studies described in these applications. Measures against bias were described at very low rates, ranging on average from 2.4% for statistical analysis plan to 19% for primary outcome variable in applications for animal experiments, and from 0.0% for sample size calculation to 34% for statistical analysis plan in publications from these experiments. Calculating an internal validity score (IVS) based on the proportion of the seven measures against bias, we found a weak positive correlation between the IVS of applications and that of publications (Spearman's rho = 0.34, p = 0.014), indicating that the rates of description of these measures in applications partly predict their rates of reporting in publications. These results indicate that the authorities licensing animal experiments are lacking important information about experimental conduct that determines the scientific validity of the findings, which may be critical for the weight attributed to the benefit of the research in the harm-benefit analysis. Similar to manuscripts getting accepted for publication despite poor reporting of measures against bias, applications for animal experiments may often be approved based on implicit confidence rather than explicit evidence of scientific rigor. Our findings shed serious doubt on the current authorization procedure for animal

PLOS BIOLOGY

**Competing Interests:** The authors have declared that no competing interests exist.

**Abbreviations:** ALS, amyotrophic lateral sclerosis; API, Animal Protection Index; ARRIVE, Animal Research: Reporting of In Vivo Experiments; AS, accuracy score; CDRP, cats, dogs, rabbits, and primates; CI, confidence interval; FSVO, Swiss Federal Food Safety and Veterinary Office; IF, impact factor; IVS, internal validity score; NC3Rs, National Centre for the Replacement, Refinement & Reduction of Animals in Research; OR, odds ratio. experiments, as well as the peer-review process for scientific publications, which in the long run may undermine the credibility of research. Developing existing authorization procedures that are already in place in many countries towards a preregistration system for animal research is one promising way to reform the system. This would not only benefit the scientific validity of findings from animal experiments but also help to avoid unnecessary harm to animals for inconclusive research.

# Author Summary

Scientific validity of research findings depends on scientific rigor, including measures to avoid bias, such as random allocation of animals to treatment groups (randomization) and assessing outcome measures without knowing to which treatment groups the animals belong (blinding). However, measures against bias are rarely reported in publications, and systematic reviews found that poor reporting was associated with larger treatment effects, suggesting bias. Here we studied whether risk of bias could be predicted from study protocols submitted for ethical review. We assessed mention of seven basic measures against bias in study protocols submitted for approval in Switzerland and in publications resulting from these studies. Measures against bias were mentioned at very low rates both in study protocols (2%-19%) and in publications (0%-34%). However, we found a weak positive correlation, indicating that the rates at which measures against bias were mentioned in study protocols predicted the rates at which they were reported in publications. Our results indicate that animal experiments are often licensed based on confidence rather than evidence of scientific rigor, which may compromise scientific validity and induce unnecessary harm to animals caused by inconclusive research.

# Introduction

Reproducibility is a fundamental principle of the scientific method and distinguishes scientific evidence from mere anecdote. The advancement of basic as well as applied research depends on the reproducibility of the findings, and can be seriously hampered if reproducibility is poor. However, accumulating evidence indicates that reproducibility is poor in many disciplines across the life sciences [1]. For example, in a study on microarray gene expression, only 8 out of 18 studies could be reproduced [2]; Prinz and colleagues [3] found large inconsistencies (65%) between published and in-house data in the fields of oncology, women's health, and cardiovascular diseases; oncologists from Amgen could confirm only 6 out of 53 published findings [4]; and, of more than 100 compounds that showed promising effects on amyotrophic lateral sclerosis (ALS) in preclinical trials, none displayed the same effect when retested by the ALS Therapy Development Institute in Cambridge [5]. Besides a waste of time and resources for inconclusive research [6–8], however, poor reproducibility also entails serious ethical problems. In clinical research, irreproducibility of preclinical research may expose patients to unnecessary risks [9,10], while in basic and preclinical animal research, it may cause unjustified harm to experimental animals [11].

Reproducibility critically depends on experimental design and conduct, which together account for the internal and external validity of experimental results [12]. External validity refers to how applicable results are to other environmental conditions, experimenters, study

populations, and even to other strains or species of animals (including humans) [12]. Thus, it also determines reproducibility of the results across replicate studies (i.e., across different labs, different experimenters, different study populations, etc.) [11,13,14]. Internal validity refers to the extent to which a causal relation between experimental treatment and outcome is warranted, and critically depends on scientific rigor, i.e., the extent to which experimental design and conduct minimize systematic bias [12,15]. It has been suggested that poor internal validity due to a lack of scientific rigor may also be a major cause of poor reproducibility in animal research [16–18].

There are various sources of bias (e.g., selection bias, performance bias, detection bias), and specific measures exist to mitigate them (e.g., randomization, blinding, sample-size calculation; [12,15,19,20]). To assess the internal validity of studies, e.g., in the peer review process, and to facilitate replication of studies, publications must contain sufficiently detailed information about experimental design and conduct, including measures taken against risks of bias [20,21]. However, systematic reviews generally found a low prevalence of reporting of measures against risks of bias (further referred to as reporting) in animal research publications. Thus, reporting ranged from 8% to 55.6% for allocation concealment, from 3% to 61% for blinded outcome assessment, from 7% to 55% for randomization, and from 0% to 3% for sample size calculation [19,22–29].

Low rates of reporting have been interpreted as evidence for a lack of scientific rigor (e.g., [20]). Indeed, several systematic reviews found correlations between poor reporting and overstated treatment effects [19,29–31]. Reporting guidelines have thus become a major weapon in the fight against risks of bias in animal research [32]. However, although the ARRIVE guidelines (Animal Research: Reporting of In Vivo Experiments) by the United Kingdom-based organization NC3Rs (National Centre for the Replacement, Refinement & Reduction of Animals in Research) have been endorsed by over 1,000 journals, this did not lead to a substantial improvement of reporting in animal studies [33]. Nevertheless, awareness seems to rise, as Macleod and colleagues [28] recently found that reporting increased over the past decades, although there is still considerable scope for improvement.

Research on the internal validity of animal experiments has focused mainly on reporting in scientific publications. However, most published research has undergone peer review when submitted for funding, and in some countries (e.g., Switzerland, Germany), individual animal experiments are licensed by national or regional authorities. For example, in Switzerland, the licensing of animal experiments is based on an explicit harm–benefit analysis, whereby any harm imposed on the animals is gauged against the expected benefit (gain of knowledge) of the experiment. Because the gain of knowledge critically depends on the scientific validity of the findings, risks of bias may affect the weight attributed to the expected benefit of a study in the harm–benefit analysis. An accurate harm–benefit analysis thus depends on information regarding risks of bias and measures used to mitigate them.

In the present study, we therefore screened applications for animal experiments submitted to the cantonal authorities in Switzerland (n = 1,277) for evidence of the use of measures to avoid risks of bias, and compared the rates at which these measures were described in applications (for reasons of simplicity hereafter also referred to as reporting) with the rates of reporting of the same measures in a representative sub-sample of publications (n = 50) resulting from experiments described in these applications. This allowed us, for the first time, to compare evidence of scientific rigor available to the authorities when licensing animal experiments with the evidence reported in scientific publications, and to assess whether poor reporting in the scientific literature is predicted by poor reporting in applications for experiments.

#### Results

Our database included a final sample of 1,277 applications for animal experiments approved by the cantonal authorities of Switzerland in the years 2008, 2010, and 2012, respectively. Evidence of scientific rigor was assessed based on seven common measures against risks of bias: allocation concealment, randomization, blinded outcome assessment, sample size calculation, inclusion and exclusion criteria, primary outcome, and a statistical analysis plan (S2 and S3 Texts). Besides analyzing each item separately, we also calculated an internal validity score (IVS; see Eq 1), which served as the primary outcome variable for the statistical analysis of effects of various study descriptors on rates of reporting. In addition, we calculated an accuracy score (AS; see Eq 2) based on six items of information explicitly asked for on the application form as a measure of how accurately the applicants had filled out the application form to control for effects of accuracy on the IVS.

### **Reporting Rates**

Reporting rates were generally very low (Table 1); on average, less than one out of the seven items were reported in applications for animal experiments, with reporting rates varying among the seven items, ranging from 2.4% for the statistical analysis plan to 18.5% for the primary outcome variable (Table 1). However, reporting rates greatly differed between individual applications, with the IVS ranging from 0 (i.e., 0/7 items reported) to 0.857 (i.e., 6/7 items reported), whereby 711 out of the 1,277 applications (55.68%) scored 0 (S1 Fig).

## Influence of Study Descriptors

We hypothesized that reporting rates and, thus, the IVS might depend on various characteristics of the studies, including the year of authorization (Year), the types of animals used (Species), the severity of the experimental procedures (Severity), the institution conducting the study (Institution), the canton authorizing the study (Canton), and the language in which the application was written (Language), as well as the AS of the application. Generalized linear models in a Bayesian information criterion selection process were used to identify which of the study descriptors best described our data, indicating that they were most likely to have influenced the IVS. The best fitting model included Year, Canton, Language, Institution, and the interaction between Species and AS (see Eq 4). According to the model output (S1 Data), however, none of the individual descriptors had a significant effect on the IVS except Language, as applications written in German had a significantly higher IVS compared to applications written in English (odds ratio [OR] = 0.79, 95% confidence interval [CI] = 0.64-0.98) and applications written in French (OR = 0.46, CI = 0.32-0.65), and the interaction between farm animals and AS (OR = 168.24, CI = 1.17-2,5571.31). Thus, below we report trends that were observed regarding effects of the descriptors that were included in the final model on the IVS.

The IVS was similar across all three years of authorization: 2012 (median = 0.0, range: 0–0.71), 2010 (0.0, 0–0.71), and 2008 (0.0, 0–0.85). At the level of individual items, trends of improvement across years were observed in the reporting rates of blinding, sample size calculation, and statistical analysis plan (Fig 1B–1E). While there was some variation in IVS across cantons, canton did not seem to have a strong effect (Fig 2). Among the different research institutions, academic institutions (i.e., universities, federal institutes of technology, or university hospitals) accounted by far for the largest part of applications, with 972 (76%) applications compared to 87 (7%) from industry, 56 (4%) from governmental institutions, and 162 (13%) from other private institutions. Overall, academic institutions (0.0, 0–0.86) tended to score lower on IVS than institutions from industry (0.14, 0–0.57), governmental institutions (0.14,



Table 1. Reporting rates (in %) of measures against risks of bias in applications for animal experiments in Switzerland depending on year of authorization, type of institution, use of genetically modified animals, authorizing canton, language of the application, species category, and degree of severity of the planned procedures.

Study descriptors		Allocation concealment	Blinded outcome assessment	Random allocation to groups	Formal sample size calculation	Inclusion/ exclusion criteria	Primary outcome	Detailed statistical analysis
	Overall	17.62	3.22	12.62	7.91	11.43	18.48	2.35
Authorization year	2008	17.52	2.40	13.74	5.59	7.45	19.68	2.39
	2010	15.22	3.33	12.08	7.09	8.51	16.55	1.42
	2012	19.85	3.77	12.21	10.46	17.15	19.25	3.14
Institution Type	Academia	18.79	3.51	10.96	7.30	9.36	16.98	1.75
	Industry	20.99	2.30	21.18	4.60	16.09	24.14	3.45
	Government	7.14	1.79	31.11	16.07	21.43	26.79	3.57
	Other	11.89	2.47	12.00	10.49	17.90	21.60	4.94
Genetically modified species	Yes	18.18	2.67	5.95	8.19	6.29	17.33	1.52
	No	17.21	3.60	17.40	7.71	15.03	19.28	2.93
Cantons	Small	13.33	3.60	15.46	12.61	18.92	23.42	5.41
	1	21.02	6.52	15.95	5.98	11.96	19.57	5.98
	2	21.02	1.05	4.49	9.38	4.69	16.67	0.00
	3	10.67	1.40	8.05	3.85	6.64	10.84	0.00
	4	21.13	5.07	14.52	9.23	13.99	19.94	1.19
	5	21.05	0.96	16.16	2.88	9.62	25.00	5.77
	6	11.54	1.56	28.07	20.31	28.13	28.13	4.69
Languages	EN	24.84	4.07	13.56	6.65	9.61	19.22	2.40
	DE	16.61	4.97	17.09	11.57	18.73	25.90	4.68
	FR	8.31	0.27	7.06	6.17	6.97	10.19	0.00
Species categories	CDRP	14.52	15.00	46.15	7.50	41.25	28.45	3.75
	Farm animal	14.40	5.11	22.22	12.50	27.27	19.32	5.11
	Other mammals	0.00	0.00	40.00	13.0	13.00	38.00	0.0
	Lab rodent	19.38	2.18	8.80	7.83	6.53	17.41	1.63
	Non- mammals	6.67	0.00	7.89	0.00	4.26	17.02	3.19
Degree of severity	0	6.67	7.56	18.95	6.72	21.85	19.33	3.36
	1	11.39	2.60	14.58	4.40	12.44	18.39	2.85
	2	21.70	2.50	11.06	10.63	10.16	17.97	2.03
	3	22.40	4.55	10.16	6.06	5.30	20.45	1.52

EN: English; DE: German; FR: French; CDRP: cats, dogs, rabbits, and primates

doi:10.1371/journal.pbio.2000598.t001

0–0.71), and other private institutions (0.14, 0–0.57; Fig 3A). At the level of individual items, similar trends were observed in the reporting rates of randomization and sample size calculation (Fig 3B–3E). There was also variation in IVS depending on the species of animals used (Fig 4A). Thus, applications for experiments on "higher" mammals (i.e., cats, dogs, rabbits, and primates [CDRP]) tended to score higher (0.17, 0–0.71) compared to experiments on farm animals (0.14, 0–0.86), other mammals (0.15, 0–0.29), laboratory rodents (0.0, 0–0.71), and non-mammals (0.0, 0–0.6), respectively. A similar trend was observed in the reporting



**Fig 1. Internal validity score of applications depending on the year of authorization.** (A) Boxplot of the IVS for the three years of authorization. The dashed blue line represents the overall mean IVS for the entire sample. The red squares represent the mean IVS for each year. (B–E) Barplots (with binomial confidence intervals) representing reporting rates per year of authorization for blinding (B), randomization (C), sample size calculation (D), and statistical analysis (E). Individual data are shown in https://figshare.com/s/bc48ed5dff9e6ebd2000 (Figure 1).

PLOS BIOLOGY

rates of blinding, randomization, sample size calculation, and statistical analysis (Fig 4B-4E). Thus, applications for experiments on CDRP as well as farm animals scored higher compared to those involving laboratory rodents and non-mammals, while data from applications for experiments involving other mammals varied widely due to the small sample size (*n* = 8).

In contrast to the IVS, the AS was generally high, with a median score of 0.8, ranging from 0.11 to 1.00. Despite the low IVS and more than half of the applications scoring 0, there was a weak but positive correlation between AS and IVS (Spearman's *rho* = 0.17, p < 0.001; Fig 5).

#### Reliability

In order to ensure reliability of the data between the two investigators (TSR, LV) as well as across time, inter-rater and intra-rater reliability tests were conducted at regular intervals. Inter-rater reliability scores (see Eq 3) of the IVS ranged from 91.4% to 97.1%, while the





**Fig 2.** Internal validity score of applications depending on the authorizing canton. Boxplot of the IVS for the six largest cantons (1–6) and the group of small cantons. The dashed blue line represents the overall mean IVS for the entire sample. The red squares represent the mean IVS for each canton or group of cantons. Individual data are shown in https://figshare.com/s/bc48ed5dff9e6ebd2000 (Figure 2).

respective intra-rater reliability scores ranged from 87.1% to 95.7% for TSR and from 94.3% to 97.1% for LV. Similarly, inter-rater reliability scores of the AS ranged from 91.3% to 96.3%, while the respective intra-rater reliability scores ranged from 87.5% to 97.5% for TSR and from 92.5% to 98.8% for LV (see S2 Data).

### Comparison between Applications and Publications

In order to relate the reporting rates obtained from applications for animal experiments to reporting rates found in the scientific literature, we selected 50 publications originating from 50 independent applications in our sample, screened them for the same seven internal validity criteria, and calculated the IVS for each publication using the same method.

Similar to what we found for applications, reporting rates in the 50 publications were generally low, albeit slightly higher than in the applications (Fig 6), resulting in a median IVS of 0.14. Reporting rates for the seven items ranged from 0% for sample size calculation to 34% for the statistical analysis plan. Again, reporting rates differed greatly between individual publications, with IVS ranging from 0 to 0.6, whereby 23 out of 50 publications (46%) scored 0.

Except for sample size calculation and the primary outcome variable, reporting rates for individual items were higher in publications than in applications (see Fig 6). Whereas IVS of applications and publications were the same in 27 cases (of which 21 scored 0), it was higher in 18 pairs (which was due to a statistical analysis plan in 12 cases) and lower in five cases. This



**Fig 3.** Internal validity score of applications depending on institutions. (A) Boxplot of the IVS for the four categories of institutions. The dashed blue line represents the overall mean IVS for the entire sample. The red squares represent the mean IVS for each category of institutions. (B–E) Barplots (with binomial confidence intervals) representing the reporting rates for each category of institutions (A: academia; G: governmental institutions; I: industry; O: other) for blinding (B), randomization (C), sample size calculation (D), and statistical analysis (E). Individual data are shown in <a href="https://figshare.com/s/bc48ed5dff9e6ebd2000">https://figshare.com/s/bc48ed5dff9e6ebd2000</a> (Figure 3).

increase was corroborated by a weak positive correlation between the IVS of applications and that of publications (Spearman's *rho* = 0.34, *p* = 0.014).

### Influence of Study Descriptors

Due to the smaller sample size, not all descriptors assessed for their effects on the IVS of applications could be analyzed here. Instead, we analyzed publication-specific descriptors, namely whether or not the journal in which the study was published had endorsed the ARRIVE guidelines and the impact factor of the journal (IF). There was no significant effect of ARRIVE on IVS (yes: median = 0.14, range: 0 to 0.57; no: median = 0, range: 0 to 0.60; p = 0.69; Fig 7A). In contrast, IF had a significant negative effect on IVS (Spearman's *rho* = -0.49, p < 0.001; Fig 7B).



**Fig 4. Internal validity score of applications depending on the species of animals.** (A) Boxplot of the IVS for the five categories of animal species (CDRP: cats, dogs, rabbits and, primates; Farm: farm animals; O\_mamm: other mammals; Rodents: laboratory rodents; N\_mamm: non-mammals). The dashed blue line represents the overall mean IVS for the entire sample. The red squares represent the mean IVS for each category of animal species. (B–E) Barplots (with binomial confidence intervals) representing the reporting rates for each category of species (C: cats, dogs, rabbits and, primates; F: farm animals; O: other mammals; R: laboratory rodents; N: non-mammals) for blinding (B), randomization (C), sample size calculation (D), and statistical analysis (E). Individual data are shown in https://figshare.com/s/bc48ed5dff9e6ebd2000 (Figure 4).

# Discussion

Based on the low reporting rates in publications of animal research and evidence suggesting that poor reporting may reflect a lack of scientific rigor [19,29–31], this study examined whether poor reporting in the scientific literature is predicted by poor reporting in applications for animal experiments, that is before the studies have actually been conducted. The study was restricted to animal experiments authorized in Switzerland for two reasons. First, Switzerland has an authorization system for animal experiments that requires detailed description of study protocols for every planned study. These study protocols form the basis of the harm–benefit analysis upon which the decision for or against authorization of individual studies is based. Second, the study was facilitated by the Swiss Federal Food Safety



Fig 5. Relationship between AS and IVS. Scatter plots of IVS in relation to AS. Individual data are shown in https://figshare.com/s/bc48ed5dff9e6ebd2000 (Figure 5).

doi:10.1371/journal.pbio.2000598.g005

and Veterinary Office (FSVO) providing access to all applications for animal experiments via their online platform (e-tierversuche) through which scientists communicate with the authorities and submit their applications for animal experiments. Such unlimited access to application forms for animal experiments is unprecedented, and it is laudable that the FSVO supported this meta-research. This kind of support has notoriously proven difficult to obtain for reasons of confidentiality, as highlighted by Chan et al. [34], with respect to clinical trial protocols for meta-research. As described in the Materials and Methods, access to the application forms was possible without violating confidentiality.

# Low Reporting Rates

We evaluated 1,277 applications for animal experiments and 50 publications derived thereof and found very low reporting rates in both applications and publications (Fig 6). Reporting rates in publications were within the range reported in previous studies (e.g., [19,20]). That reporting rates in applications were similar—even slightly lower—indicates that the authorities approving animal experiments are lacking important information about experimental conduct that may be critical for evaluating the expected benefit in a harm–benefit analysis. Risks of bias





representing reporting rates of the seven internal validity criteria (Alloc: allocation concealment; Blind: blinding; Random: randomization; Sample: sample size calculation; Inc/Exc: inclusion and exclusion criteria; Outcome: primary outcome; Stats: statistical analysis). Individual data are shown in https://figshare.com/s/bc48ed5dff9e6ebd2000 (Sample Applications and Sample Publications).

doi:10.1371/journal.pbio.2000598.g006

question the scientific validity of the results, which is a precondition for a study to achieve the expected benefit. Whether the authorities are unaware of risks of bias and measures to avoid them or whether they consider them as unimportant for the benefit of the research is unknown and warrants further study. As a result, however, animal experiments are authorized based on implicit confidence rather than explicit evidence of scientific rigor. Similarly, poor reporting in publications means that manuscripts are often accepted for publication in the absence of evidence of scientific rigor. This "trust me model" of science has been criticized before [1,35,36]. It sheds serious doubts on the current authorization procedure for animal experiments as well as the peer-review process for scientific publications, which in the long run may compromise the credibility of the research.

### Relationship between Reporting in Applications and Publications

We found a weak positive correlation between the IVS of applications and that of the corresponding publications. This suggests that the reporting of bias avoidance measures in applications predicted, at least to some extent, the reporting of such measures in publications. If this reflects a consistent relationship, asking for more detailed information on experimental





Fig 7. Internal validity score of publications depending on endorsement of the ARRIVE guidelines by the journal and the journal's impact factor. (A) IVS depending on endorsement of the ARRIVE guidelines. The dashed blue line represents the overall mean IVS for the entire sample of publications. The red squares represent the mean IVS for each group. (B) Scatter plot of the IVS depending on the impact factor of the journal. Individual data are shown in https://figshare.com/s/bc48ed5dff9e6ebd2000 (Sample Publications).

conduct in applications for animal experiments might help to promote better experimental conduct as well as better reporting in publications. Asking for more detailed information at the planning stage of the research might also reduce the danger of normative responses, whereby scientists simply satisfy the guidelines (e.g., ARRIVE) at a time when it is too late to take corrective actions on experimental conduct.

The increase in the IVS of publications compared to applications was largely due to better reporting of the statistical analysis plan (S2 Fig). This is likely due to the fact that journals (and reviewers) generally insist on a detailed description of the statistical analysis. It indicates that reporting guidelines (such as ARRIVE) could potentially increase scientific quality of animal research, if editors and reviewers helped to enforce them. However, as shown by Baker et al. [33] and confirmed by the present study (Fig 7A), this has not been the case so far; publications in journals having endorsed the ARRIVE guidelines did not score higher than publications in other journals. We also found a weak positive correlation between the accuracy of completing the application forms (AS) and the IVS. Thus, applicants who answered questions in the application form more accurately had a higher IVS. As shown by Minnerup et al. [37], this further confirms that enforcement of guidelines may be important in view of improving reporting standards.

# Effects of Study Characteristics on the IVS of Applications and Publications

In the final statistical model, language was the only descriptor having a significant effect on IVS of applications for animal experiments. Applications written in German had significantly higher IVS than applications written in English or French. Several explanations may account for this result. For example, the proportion of German native speakers may have been higher among authors of German applications; German may have been mostly used by native German speakers, while English may have been used by many non-native English speakers. Similarly, French may have been used by many non-native French speakers because, apparently, authorities in French-speaking cantons of Switzerland strongly encourage submission of applications in French (own observation). However, one might not necessarily expect language skills to affect such standardized terminology (randomization, blinding, etc.), but because these items are not explicitly asked for, applicants writing in their native language might be more likely to provide unsolicited detail. Alternatively, differences in regional policies of authorities between French- and German-speaking cantons, as well as the fact that all French applications were scored by only one experimenter (LV), may have contributed to this effect, but our data do not allow us to examine these explanations further.

Apart from language, all other explanatory variables in the final model had only weak effects on IVS that did not reach statistical significance (S1 Data). For example, there was a weak tendency for the reporting rates of blinding, sample size calculation, and statistical analysis to be higher in 2012 compared to those from previous years (Fig 1). This trend might reflect increasing awareness by both researchers and authorities of the importance of reporting, and it is consistent with recent evidence from a random sample of life sciences publications [28]. However, despite the many systematic reviews revealing flaws in experimental design and conduct since Ioannidis' seminal opinion paper [38], and the wealth of solutions that have since been proposed [2,5,32,39], little progress has been made. Like Baker et al. in 2014 [33], we did not find convincing evidence that reporting had increased from applications authorized before (2008) to those authorized after (2012) publication of the ARRIVE guidelines. Again, the main reason for this might be a lack of enforcement of these guidelines by authorities as well as journal editors. However, our sample was mostly based on studies designed and authorized before the ARRIVE guidelines became widely known. That the endorsement of the ARRIVE guidelines had no effect on the IVS of publications may thus reflect the delay in such a change taking effect.

Recent evidence indicated that industry-sponsored research is less biased than academic research [40]. We therefore predicted higher rates of reporting of measures against risks of bias in applications from private compared to academic institutions. Although there was a weak tendency for applications from academic institutions to score lower on IVS compared to governmental and private institutions, we cannot exclude random variation as the source of this trend. If true, however, it might reflect the different incentives between institutions, favoring more conservative approaches in non-academic institutions [41].

An interesting tendency was found in relation to the type of animals being used. Thus, applications for experiments on CDRP, farm animals, and other mammals had slightly higher IVS than those for experiments on lab rodents and non-mammals. CDRP and, to a lesser extent, farm animals and other mammals may benefit from the attribution of a higher moral status, e.g., because they are close relatives (primates), social partners (dogs, cats, rabbits), or otherwise elicit more compassion (farm animals, other mammals) than lab rodents (that are also considered as "pest" species) and non-mammals (mostly fish; e.g., [42,43,44]). On the one hand, this might indicate that applications are assessed more carefully when the stakes are

perceived as morally high, although it would remain unclear whether this effect is due to the applicants providing more information or to the authorities asking for more. On the other hand, IVS was low throughout, and the difference between species categories was not significant. In addition, there was no such trend with increasing degree of severity of studies. Importantly, however, the Swiss Animal Welfare Act does not provide a legal basis for such "speciesism" among vertebrates, and both authors and authorities should treat all vertebrates equally.

Finally, we found a weak but significant negative relationship between the IVS of publications and the IF of the journal in which it was published. That the journal IF does not necessarily reflect the quality of research has long been known (e.g., [45]), and a systematic review of a random sample of life sciences publications recently found no evidence for a positive relationship between IF and reporting [28]. Across the whole range of journal IF in our sample of publications, IVS of 0 clearly prevails, confirming that poor reporting of measures against risk of bias is common throughout the scientific literature.

# **General Conclusions**

According to the Animal Protection Index (API) by World Animal Protection, Switzerland (together with the United Kingdom, Austria, and New Zealand) ranked top in an international comparison of animal protection policy among 50 countries (http://api. worldanimalprotection.org/). In particular, authorization of animal experiments is based on a harm–benefit analysis, and authorization is denied if, in relation to the anticipated gain in knowledge, they inflict disproportionate harm on the animals (Article 19(4), [46]). Because the anticipated gain in knowledge critically depends on experimental design and conduct, the lack of information on measures against risks of bias in applications means that, in Switzerland, authorization of animal experiments is based on implicit confidence rather than explicit evidence of scientific rigor.

Several arguments may be held against this interpretation of our results, namely (i) that the measures against risks of bias assessed here are not important determinants of scientific validity, (ii) that they are not explicitly asked for on the application form for animal experiments, (iii) that, as the system currently works, it is not the authorities' duty to assess the scientific validity of the experiments, and (iv) that the authorities' confidence in scientific rigor is well justified. First, it is certainly the case that the authorities assess the scientific rationale underlying the proposed studies, thereby assessing several important aspects of scientific validity, although these are not specified explicitly. Also, there may be other, even more important risks of bias (e.g., use of inappropriate control group) that were not included in our evaluation. However, all seven items included here are considered as relevant measures against risks of bias that may compromise scientific validity in important ways; they have therefore been included in reporting guidelines such as the ARRIVE guidelines. Second, while it is also true that the application form does not explicitly ask for allocation concealment, randomization, blinding, and inclusion or exclusion criteria, it does ask explicitly for the primary and secondary outcome variables, sample size calculation, and a detailed statistical analysis plan. Moreover, the first example of how to describe procedures presented in the explanatory notes to the application form by the FSVO starts with "The dogs are divided randomly into 3 groups," indicating that randomization is also considered a relevant aspect of the description of procedures. Even if only those measures explicitly asked for on the application form were enforced, all applications would score IVS  $\geq$  0.42 (i.e., 3/7). Third, authorities may argue that it is the peers' duty to assess and guarantee scientific rigor, while the authorities' duties (and those of their advisory committees) should be limited to assessing the scope for applying the 3Rs

(replacement of animal experiments, reduction of animal use, and refinement of procedures) and whether the expected benefits (as declared by the applicants) outweigh the harms inflicted on the animals. However, it is important to note that not all experiments are based on project proposals that have undergone scientific peer review (e.g., most applications from the private sector), and that peer review does not seem to guarantee good scientific practice [47]. Finally, whether the authorities' implicit confidence in the scientific validity of the results of licensed experiments is justified is an empirical question. Concerns that such confidence may not be warranted is largely based on studies showing a negative relationship between reporting of measures against risks of bias and inflation of treatment effect size in preclinical studies (e.g., [19,25]). Together with accumulating evidence of poor reproducibility of in vivo research, these findings have shed doubts on the quality of experimental design and conduct. However, there is clearly a need for more research on the actual implementation of measures against risks of bias in experimental animal research. We have recently conducted an online survey amongst all Swiss animal researchers to elucidate actual implementation of the same seven measures against risks of bias assessed here. Our findings suggest that although reporting rates found in the literature tend to underestimate actual implementation of these measures, there is considerable scope for improvement [48].

Lack of scientific rigor in experimental conduct is widely considered to be an important determinant of poor reproducibility of in vivo research [16,17,18]. However, this assumption is based on the indirect evidence outlined above, and has never been tested directly. Randomization, blinding, sample size calculation, and all the other measures against risks of bias assessed here mainly affect the internal validity of experiments. Although the reproducibility of results can be affected by the internal validity of studies, reproducibility depends more on the external validity of studies [11–13]. Reproducibility may thus be enhanced mainly by using design features aimed to increase the external validity of results, such as more heterogeneous study populations, independent replicate cohorts, or multicenter study designs [14,49,50]. Thus, there is also a need for more research on the relative contribution of experimental conduct and experimental design, respectively, to the reproducibility of results.

Last, but not least, besides experimental design and experimental conduct, several other factors introduce bias into the scientific literature, in particular "hypothesizing after results are known" (HARKing, [51]), *p*-hacking [52], selective reporting [53], and publication bias [54]. The most effective way of eliminating all of these biases would be prospective registration of preclinical animal experiments similar to preregistration of clinical trials [55]. Further research is certainly needed on how to facilitate practical implementation of preregistration in the face of several contentious issues such as confidentiality, property rights, and theft of ideas. However, the authorization procedure for animal experiments already in place in Switzerland (and other countries, e.g., Germany), provides an ideal basis for implementing preregistration of animal experiments, which would not only benefit the scientific validity of results from animal experiments but also minimize unnecessary harm to animals for inconclusive research. By this, Switzerland could consolidate its position as a leader in animal protection as well as extend its leadership to scientific rigor.

### **Materials and Methods**

#### Sampling Process

Applications for animal experiments (Form A, S1 Text) were selected from an anonymized database obtained from the FSVO, containing all applications submitted in Switzerland since 1983. Access to applications archived by the FSVO was based on a contract between the FSVO and the authors of this study, which guaranteed confidentiality to the applicants. Applications

were selected based on predefined inclusion and exclusion criteria. Thus, only new applications submitted during the years 2008, 2010, and 2012 were included, of which applications related to (i) diagnosis of disease, (ii) education and training, and (iii) the protection of humans, animals, and the environment by toxicological or other safety tests required by law were excluded a priori (S3 Fig). A total of 1590 applications met these criteria and were subjected to formal screening.

### Checklist

In order to assess risks of bias in the experiments described in the applications, a checklist was elaborated (S2 Text) based on checklists used in previous studies assessing the use of measures to reduce risks of biases as reported in the published literature [19,20,56]. We restricted our checklist to items that (i) are essentially applicable to all kinds of experimental studies and (ii) can be assessed objectively without specific expertise of the research topic, and included those seven items that we encountered most often in the literature: (1) allocation concealment, (2) blinded outcome assessment, (3) randomization, (4) formal sample size calculation, (5) inclusion and exclusion criteria, (6) a primary outcome variable, and (7) a statistical analysis plan. These seven items were also used to calculate an IVS based on the number of items that were reported in the application divided by the total number of items applicable to the study (max = 7).

Internal Validity Score = number of items reported /total number of items (1)

Additional items were assessed that were, however, not included in the IVS. These included additional aspects of study conduct (blinded conduct of study, randomized conduct of study, termination criteria, references for the sample size, and general statements on statistical analysis; S2 Text).

In addition, we assessed the accuracy with which the application forms (Form A) were filled out, using items that were explicitly asked for on Form A, and for which the content to be filled in was explicitly specified in the accompanying guidelines to Form A on the FSVO webpage (https://www.blv.admin.ch/dam/blv/en/dokumente/tiere/publikationen-und-forschung/ tierversuche/erlaeuterungen-form-a.pdf.download.pdf/erlaeuterungen-form-a.pdf). Furthermore, we chose items that are relevant for the harm-benefit analysis and could be determined with high reliability. The following six items were included: (1) description and justification of the methods used (e.g., by indicating references, previous results, or results from a pilot study); (2) information about the identification of individual animals; (3) the total number of animals used, the number of treatment groups, and the number of animals per treatment group; (4) reference to a score sheet for the assessment of animal welfare; (5) the degrees of severity for all animals involved in the experiments; and (6) the fate of the animals at the end of the experiments. These six items were used to calculate an AS based on the number of items reported divided by the total number of items applicable to the study (max = 6).

Accuracy Score = number of items reported /total number of items (2)

The AS was constructed as a control measure, to control for variation in IVS induced by variation in the accuracy with which the form was filled out. Both IVS and AS were assessed by scoring whether or not the respective items were reported in any of the experiments included in an application form. Thus, a "YES" was recorded if an item was reported in at least one of the described experiments and a "NO" if an item was either not reported at all or if it was unclear. If an item was not applicable to the experiment described in the application form, "NA" was recorded (more details are given in the <u>S3 Text</u>).

# **Data Collection**

The 1590 applications were randomly allocated to two investigators (LV, TSR) for formal screening (leading to two lists of 795 applications each, one for each investigator). During screening, 94 applications were excluded because they were either incomplete or not available in the archives of the FSVO. A further 36 applications were excluded because they met one or more of the exclusion criteria reported above. This left 1,460 applications that were deemed suitable for screening. Applications written in French (n = 423) or Italian (n = 5) were screened by the investigator with better knowledge of these languages (LV), regardless of their assignment to the two investigators, while applications written in German (n = 430) or English (n = 602) were screened according to their assignments to the two investigators. Therefore, a total sample of n = 935 was screened by investigator LV while a total sample of n = 525 applications was screened by investigator TSR.

To restrict analysis to experimental in vivo studies, a further 183 applications were excluded in the course of the screening process because they referred to in vitro studies (if the animals were killed before the experimental treatment was applied; n = 106), monitoring studies (if the animals were observed in the wild; n = 28), or other exceptions (e.g., breeding studies, postmortem studies; n = 49), resulting in a final sample size of n = 1,277 applications used for analysis (see S3 Fig).

Based on information provided by the applicants on Form A and used for the annual statistics of animal use by the FSVO, we also recorded several descriptors that might influence the reporting of internal validity items; these included (i) year of authorization (2008, 2010, 2012), (ii) language (English, German, French), (iii) canton (the six largest cantons of Basel, Bern, Freiburg, Geneva, Vaud, Zurich, and the group of the remaining small cantons), (iv) type of institution (academic institutions [i.e., universities, federal institutes of technology, hospitals], industry, governmental institutions [national and cantonal], other [e.g., private institutions, foundations]), (v) animal species (laboratory rodents, higher mammals [CDRP], farm animals, other mammals, non-mammals), (vi) genetically modified animals (yes, no), and (vii) the prospective degree of severity of the planned procedures as defined by the FSVO (0, 1, 2, 3).

#### Inter-rater Reliability

Prior to the screening of the selected Form A, two pilot studies on separate applications (i.e., applications authorized in 2009) were conducted to ensure the applicability of the checklist and to ensure consistency of scoring within and between investigators. To ensure consistent scoring of applications between the two investigators, both investigators screened the same 10 applications, and discrepancies were checked at the end of the day. Inter-rater reliability (Eq 3) was assessed at regular intervals (on day 1 and then after the 100th, 300th, 500th, and 700th application on the investigators' list, respectively) by assessing the proportion of agreement between the two investigators. For this, the first five applications on each investigator's list were screened by both investigators.

Reliability = (no. of items scored - tot. no. of discrepancies)/no. of items scored \* 100) (3)

Only applications written in either German or English were used for inter-rater reliability tests. Overall, 50 applications were screened twice in the course of these inter-rater reliability tests. Inter-rater reliability never dropped below 85% (S2 Data).

#### Intra-rater Reliability

To ensure that both investigators scored applications consistently over time, samples of 10 applications were re-scored at regular intervals (after 50, 150, 350, and 550 listed applications,

respectively). In addition, each investigator conducted a final intra-rater reliability test on 10 randomly chosen applications from the whole list after completing the screening procedure. If systematic discrepancies would have occurred, the applications previously scored would have been re-scored. However, as in the case of inter-rater reliability, intra-rater reliability never dropped below 85% (S2 Data).

# Sample Size Calculation

No a priori sample size calculation was performed, as all applications were included in our sample that fulfilled the inclusion/exclusion criteria. However, once the sample size was determined, we verified that it was suitable for the planned statistical analysis (see model description below).

## Statistics

The screening data from the checklists were transferred to a tabulating program (Microsoft Excel 2010.Ink, Redmond, WA, USA) and imported into the statistical software R [57].

We used descriptive statistics to represent reporting rates for individual criteria of internal validity (allocation concealment, blinded outcome assessment, randomization, sample size calculation, inclusion and exclusion criteria, primary outcome, and statistical analysis). Furthermore, influences of relevant descriptors (year, canton, institution, and animal species) were represented graphically, with median and mean IVS of the group, and overall mean IVS.

For the statistical analysis of the overall internal validity score of applications, we used generalized linear models to evaluate the influence of the a priori stated descriptors on the internal validity score. The analyses were performed in R [57] using the built in function *glm* with a binomial error distribution to account for the data structure (primary outcome as proportions). As a first step, we compared univariate models (model with one descriptor) with an intercept-only model (modelling the intercept of the internal validity score) based on significant (p < 0.05) likelihood ratio test of the package lmtest [58] in order to identify descriptors to be included in the further modelling process. The descriptors to be retained were language, canton, species category, institutions, authorization year, and accuracy of the application. In a second step, by means of an information theoretic approach to model selection using the Bayesian Information Criterion (BIC), we identified the model that best fit our data. For an automated model selection procedure, the package MuMln [59] with the function *dredge* was used to compare all models with all possible combinations of the retained descriptors (full model included also the interaction term for species category and accuracy; see Eq 4).

glm(formula = IVS ~ Language + Canton + Species Category + Accuracy + Institution + Authorization Year + Species Category : Accuracy, family = quasibinomial, weights = maxIVS) (4)

The *dredge* function ranks all descriptor combinations according to their BIC; the model with the lowest BIC was assumed to be the one representing our data best. The final model included the following main effects (descriptors): language (3 levels), cantons (7 levels), species category (5 levels), accuracy (continuous), institution (4 levels), and authorization year (3 levels). In addition to these main effects, the candidate model included the two-way interaction between species category and accuracy (corresponds to full model, cf. Eq 4). The model parameters were retrieved after correction for over dispersion (see S1 Data).

# **Publications Sampling Process and Screening**

In order to relate the reporting rates of internal validity criteria assessed here by scoring applications for animal experiments with the reporting rates of such criteria in the published literature [19,22–28,60], we also scored a sub-sample of publications originating from studies based on applications in our study sample. These were identified by searching through grant numbers mentioned in the applications and references listed as output in the annual reports to the FSVO. For 155 applications (12.1%) we identified one or more corresponding publications. This number was reduced to 139 after excluding reviews and publications that were clearly unrelated to the study described in the applications (mismatch in animal species, general topic, or methods). This low number can be explained by the fact that studies licensed in 2012 and also many of those licensed in 2010 were not yet published, and that the search for publications had to rely on grant numbers mentioned in both applications listed in the final reports required by the authorities upon completion of licensed studies (for most studies licensed in 2012 and also many of those licensed in 2010, final reports were not yet available).

For the comparison of the internal validity scores between applications and publications, we aimed to detect a medium effect size (0.3) with a statistical power of 0.8 at a significance level of p < 0.05. Based on this, we chose a sample size of n = 50, which allowed us to detect an effect size of 0.276 (G\*power for correlations, bivariate normal model) [61]. A stratified random sampling procedure was used to select 50 publications from the 139 available publications, so as to select publications derived from a representative sample of all applications with respect to canton and type of animals used. Because this sample of publications was biased towards older applications, we compared the IVS of the sub-sample of 50 applications from which these 50 publications originated with the IVS of the entire sample of applications and found no significant difference; median IVS of the entire sample of applications (n = 1,277) was 0.0 (range 0 to 0.857), compared to 0.0 (range 0 to 0.714) for the sub-sample of applications (n = 50) from which the 50 publications were derived.

The publications were screened for reporting of internal validity criteria with a checklist containing the same seven internal validity criteria as were used for applications. The screening of all 50 publications was performed by one single investigator (LV). Publications were randomly allocated to one of the 10 d of screening (five publications per day). Days of screening were separated by two non-screening days. For the publications, descriptors were impact factor of the journal and endorsement of the ARRIVE guidelines by the journal. To determine the descriptors, the impact factor for the year of the publication as well as the ARRIVE status of the journal were assessed. If it was not possible to determine the ARRIVE status of a journal for the date of publication, given that all publications were published in 2012 or later, we used the ARRIVE status of the journal in 2015. Whether or not the ARRIVE status affected the internal validity score of publications was tested with a univariate generalized linear model (binomial error distribution), with IVS as dependent and the descriptor (endorsement of ARRIVE yes or no) as independent variables.

$$glm(formula = IVS \sim Endorsement of the ARRIVE, family = quasibinomial, weights = maxIVS)$$
(5)

Whether or not the internal validity score of publications was correlated with the impact factor of the journal was investigated using a spearman rank correlation test.

To ensure that the investigator scored the publications constantly over time, an independent person randomly chose one publication per five publications screened (i.e., one per day of screening) for an intra-rater reliability test. The chosen publication was re-screened on the second following day. The reliability (Eq 3) never dropped below the threshold of 85%.

# **Supporting Information**

**S1 Fig. Distribution of internal validity score.** Individual data are shown in <u>https://figshare.</u> com/s/bc48ed5dff9e6ebd2000 (Sample Applications).

(TIFF)

**S2 Fig. Comparison of reporting of internal validity criteria between applications and resulting publications.** AC: Allocation concealment, BL: Blinding, RA: Randomization, SS: Sample size calculation, IE: Inlcusion/exclusion criteria, PO: Primary outcome, SA: Statistical analysis. Individual data are shown in <a href="https://figshare.com/s/bc48ed5dff9e6ebd2000">https://figshare.com/s/bc48ed5dff9e6ebd2000</a> (Sample Applications and Sample Publications). (TIF)

**S3 Fig. Criteria of inclusion of the application in our study.** (TIF)

S1 Text. Form A. (PDF) S2 Text. Checklist.

(PDF)

**S3 Text. Annexe to the checklist.** (PDF)

**S1 Data. Outcome of the generalized linear model.** Output from the generalized linear model used to identify factors influencing IVS of applications. Data are presented with estimate, odds ratios, and the values for the 2.50% quartile and the 97.5% quartile. For more information about the equation we refer to Materials and Methods. (XLSX)

**S2 Data. Outcome of the reliability tests.** Outcome of the reliability tests (inter-rater reliability and intra-rater reliability). Percentage agreement and number of discrepancies are available for each item composing the IVS, as well as for the IVS. For more information about the equation we refer to Materials and Methods. (XLSX)

# Acknowledgments

The authors wish to thank Heinrich Binder, Sven Süptitz, and Michel Lehmann from the Swiss Food Safety and Veterinary Office (FSVO) for help and support with access to the applications for animal experiments. They are also grateful to the authorities from the veterinary offices of the cantons of Aargau, Luzern, Freiburg, and Basel Stadt for providing additional information about specific applications, and to the authorities of the veterinary office of the canton of Zürich for providing access to their archives. Special thanks go to Emily S. Sena for helpful advice on the scoring of measures against risks of bias and review of the study protocol, and to Beatriz Vidondo for support with statistics.

### **Author Contributions**

Conceptualization: Thomas S. Reichlin, Hanno Würbel.

Data curation: Lucile Vogt, Thomas S. Reichlin.

Formal analysis: Lucile Vogt, Thomas S. Reichlin, Christina Nathues.

Funding acquisition: Hanno Würbel.

Investigation: Lucile Vogt, Thomas S. Reichlin.

Methodology: Lucile Vogt, Thomas S. Reichlin, Christina Nathues, Hanno Würbel.

Project administration: Hanno Würbel.

Supervision: Thomas S. Reichlin, Hanno Würbel.

Validation: Thomas S. Reichlin.

Visualization: Lucile Vogt.

Writing – original draft: Lucile Vogt.

Writing – review & editing: Lucile Vogt, Thomas S. Reichlin, Christina Nathues, Hanno Würbel.

#### References

- Begley CG, Ioannidis JPA. Reproducibility in science: Improving the standard for basic and preclinical research. Circ Res [Internet]. 2015 Jan 2 [cited 2015 Jul 6]; 116(1):116–26. http://circres.ahajournals. org/content/116/1/116.long PMID: 25552691
- Ioannidis JPA, Allison DB, Ball CA, Coulibaly I, Cui X, Culhane AC, et al. Repeatability of published microarray gene expression analyses. Nat Genet [Internet]. 2009 Feb [cited 2016 Feb 16]; 41(2):149– 55. http://dx.doi.org/10.1038/ng.295 PMID: 19174838
- Prinz F, Schlange T, Asadullah K. Believe it or not: How much can we rely on published data on potential drug targets? Nat Rev Drug Discov [Internet]. 2011 Sep [cited 2013 Jun 10]; 10(9):712. <a href="http://www.ncbi.nlm.nih.gov/pubmed/21892149">http://www.ncbi.nlm.nih.gov/pubmed/21892149</a> doi: 10.1038/nrd3439-c1 PMID: 21892149
- Begley CG, Ellis LM. Drug development: Raise standards for preclinical cancer research. Nature. 2012; 483(7391):531–3. doi: 10.1038/483531a PMID: 22460880
- Perrin S. Preclinical research: Make mouse studies work. Nature [Internet]. 2014 Mar 27 [cited 2015 Nov 6]; 507(7493):423–5. http://www.nature.com/news/preclinical-research-make-mouse-studieswork-1.14913 PMID: 24678540
- Arrowsmith J. Trial watch: Phase II failures: 2008–2010. Nat Rev Drug Discov [Internet]. 2011 May [cited 2013 May 25]; 10(5):328–9. http://www.ncbi.nlm.nih.gov/pubmed/21532551
- Freedman LP, Cockburn IM, Simcoe TS. The economics of reproducibility in preclinical research. PLoS Biol [Internet]. 2015 Jun 9 [cited 2015 Jun 10]; 13(6):e1002165. http://journals.plos.org/plosbiology/ article?id=10.1371/journal.pbio.1002165 PMID: 26057340
- Glasziou P, Altman DG, Bossuyt P, Boutron I, Clarke M, Julious S, et al. Reducing waste from incomplete or unusable reports of biomedical research. Lancet [Internet]. 2014 Jan 18 [cited 2016 Jun 1]; 383 (9913):267–76. http://www.sciencedirect.com/science/article/pii/S014067361362228X doi: 10.1016/S0140-6736(13)62228-X PMID: 24411647
- Pound P, Ebrahim S, Sandercock P, Bracken MB, Roberts I. Where is the evidence that animal research benefits humans? BMJ [Internet]. 2004 Feb 28 [cited 2015 Nov 9]; 328(7438):514–7. http:// www.pubmedcentral.nih.gov/articlerender.fcgi?artid=351856&tool=pmcentrez&rendertype=abstract doi: 10.1136/bmj.328.7438.514 PMID: 14988196
- Henderson VC, Demko N, Hakala A, MacKinnon N, Federico CA, Fergusson D, et al. A meta-analysis of threats to valid clinical inference in preclinical research of sunitinib. Elife [Internet]. 2015 Jan 13 [cited 2016 Mar 31]; 4:e08351. http://elifesciences.org/content/4/e08351v1#F5 PMID: 26460544
- 11. Richter S, Garner J, Würbel H. Environmental standardization: Cure or cause of poor reproducibility in animal experiments? Nat Methods [Internet]. 2009 Apr [cited 2013 Jun 10]; 6(4):257–61. http://www.ncbi.nlm.nih.gov/pubmed/1933241 doi: 10.1038/nmeth.1312 PMID: 1933241
- Bailoo JD, Reichlin TS, Würbel H. Refinement of experimental design and conduct in laboratory animal research. ILAR J [Internet]. 2014 Jan 20 [cited 2015 Sep 2]; 55(3):383–91. <u>http://ilarjournal.oxfordjournals.org/content/55/3/383.long PMID: 25541540</u>
- 13. Würbel H. Behaviour and the standardization fallacy. Nat Genet. 2000; 26(3):263. doi: <u>10.1038/81541</u> PMID: <u>11062457</u>

- Richter S, Garner J, Auer C, Kunert J, Würbel H. Systematic variation improves reproducibility of animal experiments. Nat Methods [Internet]. 2010 Mar [cited 2013 Jun 10]; 7(3):167–8. http://www.ncbi.nlm. nih.gov/pubmed/20195246 doi: 10.1038/nmeth0310-167 PMID: 20195246
- Van der Worp HB, Howells DW, Sena ES, Porritt MJ, Rewell S, O'Collins V, et al. Can animal models of disease reliably inform human studies ? PLoS Med. 2010; 7(3):e1000245. doi: 10.1371/journal.pmed. 1000245 PMID: 20361020
- Ioannidis JPA. Contradicted and initially stronger effects in highly cited clinical research. JAMA [Internet]. 2005 Jul 13 [cited 2015 Aug 31]; 294(2):218–28. <u>http://jama.jamanetwork.com/article.aspx?</u> articleid=201218 PMID: 16014596
- Macleod M. Why animal research needs to improve. Nature. 2011; 477(7366):511. doi: 10.1038/ 477511a PMID: 21956292
- Henderson VC, Kimmelman J, Fergusson D, Grimshaw JM, Hackam DG. Threats to validity in the design and conduct of preclinical efficacy studies: A systematic review of guidelines for in vivo animal experiments. PLoS Med [Internet]. 2013 Jan 23 [cited 2016 May 19]; 10(7):e1001489. http://journals. plos.org/plosmedicine/article?id=10.1371/journal.pmed.1001489 PMID: 23935460
- Sena E, van der Worp HB, Howells DW, Macleod M. How can we improve the pre-clinical development of drugs for stroke? Trends Neurosci [Internet]. 2007 Sep [cited 2013 Jun 10]; 30(9):433–9. http://www. ncbi.nlm.nih.gov/pubmed/17765332 doi: 10.1016/j.tins.2007.06.009 PMID: 17765332
- Kilkenny C, Parsons N, Kadyszewski E, Festing MFW, Cuthill IC, Fry D, et al. Survey of the quality of experimental design, statistical analysis and reporting of research using animals. PLoS One [Internet]. 2009 Jan [cited 2013 Jun 10]; 4(11):e7824. http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid= 2779358&tool=pmcentrez&rendertype=abstract doi: 10.1371/journal.pone.0007824 PMID: 19956596
- Landis S, Amara S, Asadullah K, Austin CP, Blumenstein R, Bradley EW, et al. A call for transparent reporting to optimize the predictive value of preclinical research. Nature [Internet]. 2012 Oct 11 [cited 2013 Jun 10]; 490(7419):187–91. http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid= 3511845&tool=pmcentrez&rendertype=abstract doi: 10.1038/nature11556 PMID: 23060188
- Bebarta V, Luyten D, Heard K. Emergency medicine animal research: Does use of randomization and blinding affect the results? Acad Emerg Med [Internet]. 2003; 10(6):684–7. http://www3.interscience. wiley.com/journal/119823642/abstract PMID: 12782533
- Willmot M, Gibson C, Gray L, Murphy S, Bath P. Nitric oxide synthase inhibitors in experimental ischemic stroke and their effects on infarct size and cerebral blood flow: a systematic review. Free Radic Biol Med [Internet]. 2005 Aug 1 [cited 2015 Nov 13]; 39(3):412–25. <u>http://www.sciencedirect.com/science/article/pii/S089158490500170X PMID: 15993340</u>
- Van der Worp HB, de Haam P, Morrema E, Kalkman C. Methodological quality of animal studies on neuroprotection in focal cerebral ischaemia. J Neurol [Internet]. 2005 Sep [cited 2013 Jun 10]; 252 (9):1108–14. <u>http://www.ncbi.nlm.nih.gov/pubmed/16170651</u> doi: 10.1007/s00415-005-0802-3 PMID: 16170651
- Macleod MR, van der Worp HB, Sena E, Howells DW, Dirnagl U, Donnan GA. Evidence for the efficacy of NXY-059 in experimental focal cerebral ischaemia is confounded by study quality. Stroke [Internet].
  2008 Oct [cited 2013 Jun 10]; 39(10):2824–9. http://www.ncbi.nlm.nih.gov/pubmed/18635842 doi: 10. 1161/STROKEAHA.108.515957 PMID: 18635842
- 26. Vesterinen HM, Currie GL, Carter S, Mee S, Watzlawick R, Egan KJ, et al. Systematic review and stratified meta-analysis of the efficacy of RhoA and Rho kinase inhibitors in animal models of ischaemic stroke. Syst Rev [Internet]. 2013 Jan 20 [cited 2016 Feb 16]; 2(1):33. http://systematicreviewsjournal. biomedcentral.com/articles/10.1186/2046-4053-2-33
- McCann SK, Irvine C, Mead GE, Sena ES, Currie GL, Egan KE, et al. Efficacy of antidepressants in animal models of ischemic stroke: A systematic review and meta-analysis. Stroke [Internet]. 2014 Oct 1 [cited 2016 Feb 16]; 45(10):3055–63. http://stroke.ahajournals.org/content/45/10/3055.long PMID: 25184357
- Macleod MR, Lawson McLean A, Kyriakopoulou A, Serghiou S, de Wilde A, Sherratt N, et al. Risk of bias in reports of in vivo research: A focus for improvement. PLoS Biol [Internet]. 2015 Oct 13 [cited 2015 Oct 14]; 13(10):e1002273. http://journals.plos.org/plosbiology/article?id=10.1371/journal.pbio. 1002273 PMID: 26460723
- Rooke EDM, Vesterinen HM, Sena ES, Egan KJ, Macleod MR. Dopamine agonists in animal models of Parkinson's disease: A systematic review and meta-analysis. Parkinsonism Relat Disord [Internet]. 2011 Jun [cited 2016 Feb 16]; 17(5):313–20. http://www.sciencedirect.com/science/article/pii/ S1353802011000526 doi: 10.1016/j.parkreldis.2011.02.010 PMID: 21376651
- Macleod MR, Fisher M, O'Collins V, Sena ES, Dirnagl U, Bath PMW, et al. Reprint: Good laboratory practice: preventing introduction of bias at the bench. Int J Stroke [Internet]. 2009 Feb [cited 2016 Jun

24]; 4(1):3–5. http://www.ncbi.nlm.nih.gov/pubmed/19236488 doi: 10.1111/j.1747-4949.2009.00241.x PMID: 19236488

- Vesterinen HM, Sena ES, Ffrench-Constant C, Williams A, Chandran S, Macleod MR. Improving the translational hit of experimental treatments in multiple sclerosis. Mult Scler [Internet]. 2010 Sep [cited 2016 Jun 24]; 16(9):1044–55. http://www.ncbi.nlm.nih.gov/pubmed/20685763 doi: 10.1177/ 1352458510379612 PMID: 20685763
- McNutt M. Journals unite for reproducibility. Science (80-) [Internet]. 2014 Nov 5 [cited 2016 Feb 17]; 346(6210):679–679. http://science.sciencemag.org/content/346/6210/679.abstract
- Baker D, Lidster K, Sottomayor A, Amor S. Two years later: Journals are not yet enforcing the ARRIVE guidelines on reporting standards for pre-clinical animal studies. Eisen JA, editor. PLoS Biol [Internet]. 2014 Jan [cited 2014 Oct 7]; 12(1):e1001756. http://dx.plos.org/10.1371/journal.pbio.1001756 PMID: 24409096
- Chan A-W, Upshur R, Singh JA, Ghersi D, Chapuis F, Altman DG. Research protocols: waiving confidentiality for the greater good. BMJ [Internet]. 2006 May 6 [cited 2016 Oct 10]; 332(7549):1086–9. http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=1458595&tool=pmcentrez&rendertype= abstract doi: 10.1136/bmj.332.7549.1086 PMID: 16675819
- Nosek BA, Spies JR, Motyl M. Scientific utopia: II. Restructuring incentives and practices to promote truth over publishability. Perspect Psychol Sci [Internet]. 2012 Nov 1 [cited 2016 May 12]; 7(6):615–31. http://pps.sagepub.com/content/7/6/615.abstract PMID: 26168121
- Nosek BA, Alter G, Banks GC, Borsboom D, Bowman SD, Breckler SJ, et al. Promoting an open research culture. Science (80-) [Internet]. 2015 Jun 25 [cited 2015 Jun 25]; 348(6242):1422–5. http:// www.pubmedcentral.nih.gov/articlerender.fcgi?artid=4550299&tool=pmcentrez&rendertype=abstract
- Minnerup J, Zentsch V, Schmidt A, Fisher M, Schäbitz W-R. Methodological Quality of Experimental Stroke Studies Published in the Stroke Journal: Time Trends and Effect of the Basic Science Checklist. Stroke [Internet]. 2016 Jan 1 [cited 2016 Oct 10]; 47(1):267–72. http://stroke.ahajournals.org/content/ 47/1/267.long PMID: 26658439
- Ioannidis JPA. Why most published research findings are false. PLoS Med [Internet]. 2005 Aug [cited 2013 Jun 10]; 2(8):e124. http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=1182327&tool= pmcentrez&rendertype=abstract doi: 10.1371/journal.pmed.0020124 PMID: 16060722
- Van Luijk J, Bakker B, Rovers MM, Ritskes-Hoitinga M, de Vries RBM, Leenaars M. Systematic reviews of animal studies; missing link in translational research? PLoS One [Internet]. 2014 Jan 26 [cited 2015 Oct 29]; 9(3):e89981. <u>http://journals.plos.org/plosone/article?id=10.1371/journal.pone.0089981</u> PMID: 24670965
- 40. Krauth D, Anglemyer A, Philipps R, Bero L. Nonindustry-sponsored preclinical studies on statins yield greater efficacy estimates than industry-sponsored studies: A meta-analysis. PLoS Biol [Internet]. 2014 Jan 21 [cited 2016 May 11]; 12(1):e1001770. http://journals.plos.org/plosbiology/article?id=10.1371/journal.pbio.1001770 PMID: 24465178
- Macleod M. Some salt with your statin, professor? PLoS Biol [Internet]. 2014 Jan 21 [cited 2015 Aug 12]; 12(1):e1001768. http://journals.plos.org/plosbiology/article?id=10.1371/journal.pbio.1001768 PMID: 24465176
- 42. Herzog HA. The moral status of mice. ILAR J [Internet]. 1989 Jan 1 [cited 2016 Jun 24]; 31(1):4–7. http://ilarjournal.oxfordjournals.org/content/31/1/4.full
- Plous S. Psychological mechanisms in the human use of animals. J Soc Issues [Internet]. 1993 Apr [cited 2016 Jun 24]; 49(1):11–52. http://doi.wiley.com/10.1111/j.1540-4560.1993.tb00907.x
- 44. Batt S. Human attitudes towards animals in relation to species similarity to humans: A multivariate approach. Biosci Horizons [Internet]. 2009 Apr 21 [cited 2016 Jun 24]; 2(2):180–90. http://biohorizons.oxfordjournals.org/content/2/2/180.full
- **45.** Seglen PO. Why the impact factor of journals should not be used for evaluating research. BMJ [Internet]. 1997 Feb 15 [cited 2015 Oct 1]; 314(7079):498–502. http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=2126010&tool=pmcentrez&rendertype=abstract PMID: 9056804
- AWA. Animal Welfare Act [Internet]. SR 455 Die Bundesversammlung der Schweizerischen Eidgenossenschaft.; 2005. http://www.admin.ch/opc/de/classified-compilation/20022103/201405010000/455. pdf
- Curry S. Peer review, preprints and the speed of science. The Guardian [Internet]. 2015 Sep 7 [cited 2016 May 11]. https://www.theguardian.com/science/occams-corner/2015/sep/07/peer-reviewpreprints-speed-science-journals
- Reichlin TS, Vogt L, Würbel H. The researchers' view—Survey on the design, conduct, and reporting of in vivo research. PLoS ONE. 2016; 11(12):e0165999.

- 49. Richter S, Garner J, Zipser B, Lewejohann L, Sachser N, Touma C, et al. Effect of population heterogenization on the reproducibility of mouse behavior: A multi-laboratory study. PLoS One [Internet]. 2011 Jan [cited 2013 Jun 10]; 6(1):e16461. http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid= 3031565&tool=pmcentrez&rendertype=abstract doi: 10.1371/journal.pone.0016461 PMID: 21305027
- Voelkl B, Würbel H. Reproducibility crisis: Are we ignoring reaction norms? Trends Pharmacol Sci [Internet]. 2016 May 19 [cited 2016 Jun 1]; 37(7):509–10. http://www.sciencedirect.com/science/article/ pii/S0165614716300347 doi: 10.1016/j.tips.2016.05.003 PMID: 27211784
- Kerr NL. HARKing: Hypothesizing after the results are known. Personal Soc Psychol Rev [Internet]. 1998 Jan 1 [cited 2016 May 11]; 2(3):196–217. http://psr.sagepub.com/content/2/3/196.abstract
- 52. Head ML, Holman L, Lanfear R, Kahn AT, Jennions MD. The extent and consequences of P-Hacking in science. PLoS Biol [Internet]. 2015 Mar 13 [cited 2015 Mar 14]; 13(3):e1002106. http://journals.plos. org/plosbiology/article?id=10.1371/journal.pbio.1002106 PMID: 25768323
- 53. Chan A-W, Hróbjartsson A, Haahr MT, Gøtzsche PC, Altman DG. Empirical evidence for selective reporting of outcomes in randomized trials: Comparison of protocols to published articles. JAMA [Internet]. 2004 May 26 [cited 2016 May 11]; 291(20):2457–65. http://jama.jamanetwork.com/article.aspx? articleid=198809&resultclick=1 PMID: 15161896
- 54. Sena ES, van der Worp HB, Bath PMW, Howells DW, Macleod MR. Publication bias in reports of animal stroke studies leads to major overstatement of efficacy. PLoS Biol [Internet]. 2010 Mar [cited 2013 Jun 10]; 8(3):e1000344. http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=2846857&tool=pmcentrez&rendertype=abstract doi: 10.1371/journal.pbio.1000344 PMID: 20361022
- 55. Jansen of Lorkeers SJ, Doevendans PA, Chamuleau SAJ. All preclinical trials should be registered in advance in an online registry. Eur J Clin Invest [Internet]. 2014 Sep [cited 2016 Mar 1]; 44(9):891–2. http://www.ncbi.nlm.nih.gov/pubmed/25041644 doi: 10.1111/eci.12299 PMID: 25041644
- 56. Kilkenny C, Browne WJ, Cuthill IC, Emerson M, Altman DG. Improving bioscience research reporting: The ARRIVE guidelines for reporting animal research. PLoS Biol [Internet]. 2010 Jan [cited 2013 Jun 10]; 8(6):e1000412. http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=2893951&tool= pmcentrez&rendertype=abstract doi: 10.1371/journal.pbio.1000412 PMID: 20613859
- 57. R Development Core Team. R: A Language and environment for statistical computing [Internet]. 2008 [cited 2016 Mar 2]. http://www.r-project.org.
- Zeileis A, Hothorn T. Diagnostic checking in regression relationships. CRAN R News [Internet]. 2002 [cited 2016 Mar 2]; 2(3):7–10. https://cran.r-project.org/doc/Rnews/
- Barton K. MuMIn: Multi-model inference. 2015 [cited 2016 Mar 2]. https://cran.r-project.org/web/ packages/MuMIn/index.html
- 60. Willmot M, Gray L, Gibson C, Murphy S, Bath PMW. A systematic review of nitric oxide donors and larginine in experimental stroke; effects on infarct size and cerebral blood flow. Nitric Oxide [Internet]. 2005 May [cited 2015 Nov 13]; 12(3):141–9. http://www.sciencedirect.com/science/article/pii/ S1089860305000042 PMID: 15797842
- Faul F, Erdfelder E, Buchner A, Lang A-G. Statistical power analyses using G\*Power 3.1: Tests for correlation and regression analyses. Behav Res Methods [Internet]. 2009 Nov [cited 2014 Jul 9]; 41 (4):1149–60. http://www.ncbi.nlm.nih.gov/pubmed/19897823 doi: 10.3758/BRM.41.4.1149 PMID: 19897823